Author's response to reviews

Title: The effect of the stay active advice on physical activity and on the course of acute severe low back pain.

Authors:

  Patricia Olaya-Contreras (polayac@health.sdu.dk)
  Jorma Styf (jorma.styf@orthop.gu.se)
  Daniel Arvidsson (daniel.arvidsson@med.lu.se)
  Karin Frennered (karin.frennered@vgregion.se)
  Tommy Hansson (tommy.hansson@orthop.gu.se)

Version: 6  Date: 7 August 2015

Author's response to reviews: see over
Cover letter

Dear editor
Below we provide our responses to your comments and we hope that they will give sufficient information to satisfy your request.

Comment 1
TRN - We notice that you are reporting a clinical trial but have not cited a trial registration number. This must be obtained before we can proceed to your manuscript.

Response 1
We have now performed trial registration at ClinicalTrials.gov and have received the trial number (NCT02517762). This information is now included at the end of the abstract and in the first paragraph of the method section.

Yours sincerely,

The authors